Author's response to reviews

Title: What do adolescents want in order to become more active?

Authors:

Kirsten Corder (klc29@medschl.cam.ac.uk)
Andrew J Atkin (aa595@medschl.cam.ac.uk)
Ulf Ekelund (ulf.ekelund@nih.no)
Esther MF van Sluijs (ev234@medschl.cam.ac.uk)

Version: 3 Date: 22 July 2013

Author's response to reviews:

Kirsten Corder, PhD
MRC Epidemiology Unit, University of Cambridge UK
Tel: 01223 769150
Fax: 01223 330316
klc29@medschl.cam.ac.uk

The Editor
BMC Public Health

July 22nd, 2013

Dear Editor,

This letter accompanies the resubmission of a manuscript “MS: 9744708149712317 What do adolescents want in order to become more active?” by Kirsten Corder and co-authors that is being submitted for possible publication in BMC Public Health.

The authors thank the editors for the opportunity to resubmit this manuscript.

A point by point response to the comments is included below and the changes in the revised version of the manuscript are highlighted in blue.

Yours sincerely,

Kirsten Corder, PhD
MRC Epidemiology Unit, University of Cambridge UK

Point by point response to comments

The authors thank the reviewers for their positive reviews of this manuscript.

Minor Essential Revisions:
1. Results
- Enumerate correctly the table 4 in the title of the table (it is currently enumerated as table 3)

The authors have corrected this error.